Author's response to reviews

Title: Allergic Bronchopulmonary Aspergillosis with coexistent Aspergilloma: a 5-year follow-up

Authors:

Izidor Kern (izidor.kern@klinika-golnik.si)
Anton Lopert (Anton.Lopert@guest.arnes.si)

Version: 3 Date: 29 July 2009

Author's response to reviews:

1 - Referee 1 suggested additional reference. Shah et al. published the existence of ABPA and allergic Aspergillus sinusitis for the first time in 1996. Their case is cited in our manuscript as a reference 9. Suggested reference from 2006 does not have any new information.

2 - Referee 2 made comments regarding English. The writing of manuscript was corrected and reviewed by English speaker. Figure 1 shows lung infiltrate with a cavity and is best observed on a chest radiograph. Bronchiectasis are illustrated by CT scan on Figure 4. In the text, we described that patient was diagnosed having bronchiectasis based on CT scan.